Author's response to reviews

Title: A standard procedure for creating a frailty index

Authors:

Samuel D Searle (ssearle@dal.ca)
Arnold Mitnitski (arnold.mitnitski@dal.ca)
Evelyne A Gahbauer (evelyne.gahbauer@yale.edu)
Thomas M Gill (thomas.gill@yale.edu)
Kenneth Rockwood (kenneth.rockwood@dal.ca)

Version: 5 Date: 17 September 2008

Author's response to reviews:

REVISIONS to Editor: A standard procedure for creating a frailty index

Cover Page:

Email address to be lowercase for the following:

arnold.mitnitski@dal.ca

kenneth.rockwood@dal.ca

Page 6.

Second last sentence in first paragraph: Mortality was checked monthly.

Second paragraph – third sentence: data dictionary. Remove space and extra period.

Page 13.
2nd paragraph last sentence. multivariable analysis, all these variables. – all these.

Page 18

Acknowledgements:

Third sentence: Fountain Innovation Fund of the Queen Elizabeth II Health Sciences Foundation. (remove extra period and space). Dr.

Page 19: References: The following titles have been changed to be lowercase to be consistent with the other references. Also a space correction in number 4.


Figure 1. Is should read Number of People. It currently reads People.